Author's response to reviews

Title: Antioxidant-rich dietary intervention for improving asthma control in pregnancies complicated by asthma: study protocol for a randomized controlled trial.

Authors:

Jessica A Grieger Dr (jessica.grieger@adelaide.edu.au)
Lisa G Wood Assoc/Prof (Lisa.Wood@newcastle.edu.au)
Vicki L Clifton Assoc/Proff (vicki.clifton@adelaide.edu.au)

Version: 2 Date: 28 August 2013

Author's response to reviews:

We have amended table 1 as the "tick" marks did not show up in the original version. We have now used "crosses" to show where study endpoints will be taken.